Author's response to reviews

Title: Does the aldosterone: renin ratio predict the efficacy of spironolactone over bendroflumethiazide in hypertension? A clinical trial protocol for RENALDO (RENNi-ALDOIsterone) study

Authors:

Hari K Parthasarathy (krishnankala@hotmail.com)
Khamis Alhashmi (alhashmi1@yahoo.com)
Alex McMahon (alexm@stats.gla.ac.uk)
Allan D Struthers (a.d.struthers@dundee.ac.uk)
John MC Connell (mcc1m@clinmed.gla.ac.uk)
Gordon T McInnes (g.t.mcinnes@clinmed.gla.ac.uk)
Ian Ford (ian@stats.gla.ac.uk)
Thomas M MacDonald (t.m.macdonald@dundee.ac.uk)

Version: 4 Date: 3 May 2007

Author's response to reviews: see over
The changes made to the previous version are

1. Title aligned to the left
2. Author names changed to the format as in the submission system
3. Country added in affiliation
4. Abbreviations title and content removed
5. Figure title removed
6. Figures cropped and changed to JPEG
7. Funding info added to the acknowledgements
8. Full author list included in the references if up to 30 names and et al removed where relevant